Author's response to reviews

Title: Design of the Anti-tuberculosis Drugs induced Adverse Reactions in China National Tuberculosis Prevention and Control Scheme Study (ADACS)

Authors:

Yin Yin Xia (yinyin_xia@yahoo.cn)
Dai Yu Hu (hukaixiou@263.net)
Fei Ying Liu (liufeijing@163.com)
Xiao Meng Wang (xmwang@cdc.zj.cn)
Yan Li Yuan (yuanyanli7500@163.com)
De Hua Tu (bata0405@126.com)
Yi Xin Chen (yix_chen@263.net)
Lin Zhou (zhoulin@chinatb.org)
Li Zhen Zhu (zhulizhen@hotmail.com)
Wei Wei Gao (gwwiys@sina.com)
Hong Yuan Wang (why99@vip.sina.com)
Da Fang Chen (dafangchen@bjmu.edu.cn)
Li Yang (lyang@bjmu.edu.cn)
Ping Ping He (pingpinghe@vip.163.com)
Xiao Ting Li (imissfriend@163.com)
Ying Jian He (mimi487@163.com)
Feng Sun (sunfeng801003@sina.com)
Si Yan Zhan (siyan-zhan@bjmu.edu.cn)

Version: 3 Date: 6 May 2010

Author's response to reviews:

Natalie Pafitis
BMC Public Health
May 6th, 2010

Dear Natalie:

Thank you for your review opinion. Please find our revised manuscript. We rewrote the sampling calculation part of our study in page 7 and 8, making a specific description of steps we took for total sample size assumption. If something remains unclear, please inform me.

Best regards.

Yours sincerely,

SiYan Zhan